Author's response to reviews

Title: Quality of care of patients with acute myocardial infarction in Bulgaria: a cross-sectional study

Authors:

   Milka Ganova-Iolovska (ganova_milka@yahoo.com)
   Krassimir Kalinov (kkalinov@medistat-bg.com)
   Max Geraedts (geraedts@uni-duesseldorf.de)

Version: 4 Date: 7 January 2009

Author's response to reviews:

7 January 2009

To the Editorial Board – Health Services Research:

We thank you and the reviewers very much for your helpful comments and suggestions. We have now incorporated most of the recommendations. In the following, we will address (a) the main concerns you have summarized and (b) each of the further concerns.

Detailed suggestions of reviewer (Mr. Glickman):

(1) Table 4: Reviewer (Mr. Glickman) recommended including more information concerning the association of the variables age and education level with pre-hospital time delay. We accepted this recommendation and included additional information in Table 4, and in section “Results”, paragraph “Pre-hospital time delay”, as well as in section “Discussion”, paragraph “Pre-hospital time delay”. Furthermore, we included two additional references related to the topic.

(2) Additionally, we added information on the tests used to analyze our hypotheses.

We hope that our revision addressed all comments appropriately and look forward to your decision.

Sincerely,
Max Geraedts